Author's response to reviews

Title: Increased permeability-oedema versus atelectasis in pulmonary dysfunction after trauma and surgery: a prospective cohort study

Authors:

A.B. Johan Groeneveld (johan.groeneveld@vumc.nl)

Version: 4 Date: 6 July 2007

Author's response to reviews:

Revisions as requested per email